DETERMINANTS OF BANK PROFITABILITY: EVIDENCE FROM THE EMERGING ECONOMY

Tafirei Mashamba^{*}, Chisinga N. Chikutuma^{*}

 * Corresponding author, College of Accounting Sciences, University of South Africa, Pretoria, South Africa; School of Law and Business Sciences, Great Zimbabwe University, Masvingo, Zimbabwe
 Contact details: College of Accounting Sciences, University of South Africa, 1 Preller Street, Muckleneuk Ridge, Pretoria, South Africa
 ** College of Accounting Sciences, University of South Africa, Pretoria, South Africa

How to cite this paper: Mashamba, T., & Chikutuma, C. N. (2023). Determinants of bank profitability: Evidence from the emerging economy [Special issue]. *Corporate & Business Strategy Review*, 4(4), 310–323.

https://doi.org/10.22495/cbsrv4i4siart12

Copyright © 2023 The Authors

This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0). https://creativecommons.org/licenses/by /4.0/

ISSN Online: 2708-4965 ISSN Print: 2708-9924

Received: 28.01.2023 Accepted: 30.11.2023

JEL Classification: C23, E44, G21, O16 **DOI:** 10.22495/cbsrv4i4siart12

Abstract

This study examines the determinants of bank profitability in Zimbabwe, a country that has faced severe economic challenges in the past decade. To address this issue, we use a panel data analysis of 11 commercial banks over the period 2011-2020 and apply the system generalised method of moments (GMM) estimator to control heterogeneity and endogeneity issues. We find that bank-specific factors, such as non-interest income, liquidity, cost efficiency, capital adequacy, and bank stability, have a positive and significant impact on bank profitability, while the industry factor, bank concentration, has a negative and significant impact on bank profitability. We also find that macroeconomic factors, such as gross domestic product (GDP) and inflation, do not have a significant influence on bank profitability. This result is surprising given the high inflation and low growth rates experienced by Zimbabwe in recent years. Moreover, we find that regulatory capital weakens the positive effect of bank stability on bank profitability in Zimbabwe. This result suggests that higher capital requirements may reduce the risk-taking incentives or opportunities of banks, which may lower their profitability potential. Finally, we find no evidence of a moderating effect of fintech on bank performance. This result implies that fintech may not significantly impact the competitiveness and performance of banks in Zimbabwe in non-interest income activities. Our study concludes that bank profitability in Zimbabwe is mainly determined by internal factors that are under the control of bank managers and regulators, rather than external factors that are beyond their influence. Based on these findings, we provide several policy implications and recommendations for enhancing bank profitability and fostering a sound and resilient banking sector in Zimbabwe.

Keywords: Banks, Profitability, Panel Data, System GMM, Zimbabwe

Authors' individual contribution: Conceptualization — T.M. and C.N.C.; Methodology — T.M.; Software — T.M.; Formal Analysis — T.M.; Investigation — T.M. and C.N.C.; Resources — C.N.C.; Data Curation — T.M.; Writing — Original Draft — T.M.; Writing — Writing & Editing — T.M.; Supervision — C.N.C.; Project Administration — C.N.C.; Funding — C.N.C.

Declaration of conflicting interests: The Authors declare that there is no conflict of interest.

VIRTUS 310

1. INTRODUCTION

Banks are essential for the functioning of the financial system and the economy as a whole, as they perform the role of intermediaries between savers and investors and facilitate the allocation of funds to productive activities and economic growth. However, to carry out this crucial service, banks need to be profitable and stable. The main source of income for commercial banks is the interest they charge on loans (Dang & Nguyen, 2022; DeYoung & Rice, 2004). This implies that bank profits tend to vary with the business cycle, increasing in periods of expansion and decreasing in periods of contraction (Satria et al., 2016). There are two main reasons for this procyclicality. First, bank lending is influenced by economic output (gross domestic product, GDP). When the economy is in a downturn, banks reduce their credit supply due to low demand for loans from firms and households that face limited investment opportunities (Satria et al., 2016; Berlin, 2012). This relationship has been confirmed by several studies (Flamini et al., 2009; Athanasoglou et al., 2008). Second, loan losses tend to rise in times of recession due to higher default rates from borrowers who suffer from business losses and income shocks, which require banks to increase their provisions (Muriu, 2023; Albertazzi & Gambacorta, 2009). Moreover, the negative correlation between economic output and the probability of loan default is asymmetric, meaning that it is stronger during recessions than during expansions (Bolt et al., 2012).

However, the case of Zimbabwe presents a unique challenge to this conventional wisdom. Over the past few years, Zimbabwe has experienced a significant decline in its GDP, with the exception of a slight increase in 2017 attributed to favourable rainfall (International Monetary Fund [IMF], 2022). Surprisingly, amidst this overall economic decline, the profitability of commercial banks in the country has shown consistent improvement. This raises the question of what factors are contributing to the impressive performance of banks in Zimbabwe, despite the numerous challenges they face. One major obstacle facing banks in Zimbabwe is currency instability. The country has experienced significant fluctuations in its currency (Mahonye & Zengeni, 2019), which poses risks to the financial sector. Additionally, policy uncertainty further compounds the challenges faced by banks, as constantly changing regulations and economic policies can make it difficult for businesses, banks included, to operate effectively and plan for the future (Maumbe & Chikoko, 2022).

Furthermore, banks in Zimbabwe face intense competition from mobile money providers. These mobile banking platforms have gained significant popularity in the country due to their convenience and accessibility (Mutsonziwa & Maposa, 2016). This competition puts pressure on traditional banks to innovate and provide additional value to customers in order to maintain their market share and political profitability. Moreover, turmoil in Zimbabwe has added another layer of complexity to the banking industry. Political instability can create an uncertain business environment (Maune, 2015), affecting investor confidence and economic growth (Komal & Abbas, 2015). Banks must navigate this challenging political landscape and adapt their strategies accordingly to ensure their continued success.

Despite these obstacles, the banks in Zimbabwe have continued to post impressive results. This suggests that there may be other factors at play contributing to their success. Thus, Zimbabwe presents an interesting case study that challenges the conventional wisdom that economic decline directly translates to poor performance for banks. facing currency instability, Despite policy uncertainty, intense competition, and political turmoil, banks in Zimbabwe have managed to thrive. Understanding the factors behind their success requires further analysis of their risk management, operational strategies, and ability to innovate in response to the evolving market dynamics. Thus, this paper aims to contribute to the existing literature on bank profitability by examining the unique case of Zimbabwe, a developing economy with distinct characteristics.

In addition to examining the main factors that influence bank profitability in Zimbabwe, this paper also explores the moderating effects of bank and fintech on this relationship. stability The moderating analysis aims to test whether bank stability and fintech enhance or weaken the impact of the determinants of bank profitability on bank profitability. For example, it is possible that bank stability strengthens the positive effect of capital adequacy on profitability, as more stable banks may face lower funding costs and higher market confidence. On the other hand, it is possible that fintech weakens the positive effect of loan market share on profitability, as more digital financial services may reduce the demand for traditional bank loans. The moderating analysis provides insights into how banks can leverage their stability and innovation to improve their profitability and a challenging performance in environment. By examining these issues, this study aims to provide insights and evidence regarding the determinants of bank profitability in Zimbabwe, helping policymakers and regulators develop effective strategies to ensure bank stability and efficient regulation. Furthermore, as bank profitability has significant implications for the viability and sustainability of banks and their intermediation function, the findings of this study can contribute to the overall financial well-being and economic growth of Zimbabwe.

This paper is organized as follows. Section 2 reviews the relevant literature on the determinants of bank profitability. Section 3 describes the data, variables, and methodology used in the empirical analysis. Section 4 reports and discusses the main findings and results of the panel data regression. Section 5 summarizes the key conclusions, policy implications, and recommendations of the study.

2. LITERATURE REVIEW

This literature review section surveys the existing literature that has investigated the determinants of bank profitability in both developed and emerging economies. It focuses on discussing the key variables that affect bank profitability and providing consistent and conflicting evidence from previous studies.

VIRTUS 311

The relationship between bank liquidity and profitability has been extensively studied, but there are conflicting views in the literature. Pracoyo and Imani (2018) found that banks targeting low levels of liquid assets tend to be more profitable. This finding is supported by Bordeleau and Graham (2010), who found that higher levels of liquid assets lead to greater profitability by reducing financing costs. On the other hand, Abbas et al. (2019) argue that holding more liquid assets has an opportunity cost in terms of foregone interest income. Tran et al. (2016) and Goddard et al. (2013) also find a negative relationship between liquidity and bank profitability.

The relationship between bank capital and profitability has also been investigated extensively, but the results are mixed. Abbas et al. (2019) found a positive association between bank capital and profitability during the post-crisis era in Asia and the United States. Ozili (2017) also found a positive influence of bank capital on the profitability of commercial banks in Africa. However, Berger and Bouwman (2013) and Barth et al. (2008) concluded that the effect of capital on bank profitability is indeterminate. Tran et al. (2016) reported mixed results, finding an inverse relationship between bank capital and profitability for larger banks, and a positive relationship for smaller institutions. This lack of consensus on the relationship between bank capital and profitability indicates a need for further research.

The impact of credit risk on bank profitability remains inconclusive as well. On one hand, higher credit risk is associated with higher profit, as found by Tarus et al. (2012). On the contrary, bank profitability is negatively affected if scheduled loan payments are not collected. Studies such as Saleh and Afifa (2020), Abbas et al. (2019), Islam and Nishiyama (2016), and Ozili (2015) report an inverse relationship between credit risk and bank profitability, suggesting that suboptimal lending quality leads to high loan loss provisions and lower overall profitability. The impact of credit risk on bank profitability thus remains inconclusive.

Turning to operational efficiency, Karakaya and Er (2013) report that larger banks have higher overheads compared to smaller banks. Fungáčová et al. (2020) attribute this lower efficiency to structural problems and political incentives that prevent cost minimization, thereby adversely impacting bank profitability. However, Singh (2021) argues that the utilization of new financial technologies, such as automatic teller machines (ATMs) and the Internet, has led to a decrease in overhead expenses and an increase in profitability. A cross-country analysis by Le and Ngo (2020) supports this idea, demonstrating that the growth in bank cards issued, ATMs, and POS terminals enhances bank profitability.

Nonetheless, many studies emphasize the relationship between bank profitability and the macroeconomic environment. For instance, Yüksel et al. (2018) employ the generalised method of moments (GMM) method to examine bank profitability in 13 post-Soviet countries and find that higher GDP is associated with higher profitability. Similarly, Deng (2016) and Al-Jafari and Alchami (2014) corroborate the positive impact of GDP growth on bank profitability. However, Simiyu and Ngile (2015), and Tan and Floros (2012) offer different perspectives, finding insignificant, positive but insignificant, and negative effects of GDP on bank profitability, respectively.

The existing literature on the relationship between inflation and bank profitability also yields mixed results. Several studies, including those by Athanasoglou et al. (2008) and Demirgüç-Kunt and Huizinga (1999) have argued for a positive impact of inflation on bank profitability, based on their research conducted in Greece, the Eurozone, and Middle Eastern Islamic banks, respectively. However, alternative studies conducted by Scott and Ovuefeyen (2014) and Moyo and Tursoy (2020) found a significant negative relationship between profitability and banks' the inflation rate In contrast, other studies by Combey and Togbenou (2017), and Ally (2014) have established that inflation has no impact on the profitability of commercial banks. This divergence of findings in the literature highlights the lack of consensus regarding the determinants of bank profitability. To contribute to this ongoing discourse, the current study seeks to provide empirical evidence from Zimbabwe, a developing Sub-Saharan African economy.

3. DATA AND METHODOLOGY

3.1. Data

This study examines the determinants of bank profitability in Zimbabwe over the period from 2011 to 2020. The data were collected from three sources: 1) Bureau van Dijk's Orbis database, which provides financial information on banks and other firms; 2) the World Bank's Global Financial Development Database, which provides indicators of financial sector development and performance; and 3) the World Bank's Economic Indicators for Zimbabwe, which provide data on key economic indicators such as GDP growth, inflation, and exchange rates. The population comprises of thirteen commercial banks, but the study omitted the banks that have had no data for three years or more. Thus, the sample is sorely based on data availability. This resulted in an unbalanced panel sample of eleven commercial banks¹.

3.2. Estimation approach

To estimate the determinants of bank profitability, the study adopts the following linear regression model:

$$\pi_{it} = \alpha + \beta X_{it} + \rho IND_{ct} + \vartheta MACECO_{ct} + \nu_{it} + \varepsilon_{it} \quad (1)$$

where, π_{it} denotes bank profitability measured by net interest margin (NIM); α represents the constant coefficient; X_{it} denotes a set of bank-specific characteristics; IND_{ct} denotes industry factors; $MACECO_{ct}$ denotes a set of macroeconomic factors; β , ρ , and ϑ are the coefficients for bank-specific characteristics, industry, and macroeconomic factors, respectively; v_{it} captures unobservable timeinvariant bank fixed effects that control for

¹ The list of banks used in the study includes: AFC, CABS, CBZ, Ecobank, FBC, First Capital, Nedbank, NMB, Stanbic, Standard Chartered, Steward.

heterogeneity across banks; and ε_{it} represents the error term.

Several studies have documented that bank profits tend to exhibit persistence due to market structure imperfections, such as entry barriers, market concentration, or regulatory restrictions (Gugler & Peev, 2018; Mashamba, 2018; Flamini et al., 2009). To account for this persistency, the study adopts a dynamic error component panel regression model that incorporates the lagged dependent variable as one of the explanatory variables. This model allows for both time-invariant and time-varying unobserved heterogeneity across banks as follows:

$$\pi_{it} = \alpha + \lambda \pi_{i,t-1} + \beta X_{it} + \rho I N D_{ct} + + \vartheta M A CECO_{ct} + \nu_{it} + \varepsilon_{it}$$
(2)

where, λ represents the speed of adjustment coefficient that measures the degree of mean reversion of bank profitability; and the other variables are as defined previously. The speed of adjustment coefficient (λ) reflects the market structure and competition in the banking sector. A lower value of λ indicates a higher speed of adjustment, which implies that bank profitability is more responsive to changes in market conditions and more competitive. Conversely, a higher value of λ indicates a lower speed of adjustment, which implies that bank profitability is more rigid and less competitive. Flamini et al. (2009) found that bank profitability in Africa exhibits a low degree of mean reversion, indicating low competition and high market power.

3.3. Variables description

3.3.1. Dependent variable (NIM)

There are three main profitability metrics mainly used in literature, i.e., return on assets (ROA), return on equity (ROE), and net interest margin (NIM). Similar to Dietrich and Wanzenried (2011), Flamini et al. (2009), and Athanasoglou et al. (2008), this study uses NIM to assess bank profitability instead of ROA, which is commonly used in literature, because NIM is more sensitive to the interest rate environment, i.e., it reflects the ability of banks to adjust their interest income and expenses to changes in market conditions. The interest rate component can significantly affect NIM; hence, NIM is considered to be a more focused metric. NIM shows the profits generated from interest-bearing assets and liabilities by management.

3.3.2. Explanatory variables: Bank specific factors

Lagged dependent variable (nim_{t-1}) . As mentioned earlier, bank profits are presumed to be persistent. To capture persistence in bank profits, the variable *lagged return on assets* $(nim_{i,t-1})$ is included among the explanatory variables.

Income diversification (non-interest income, nii). a tight competitive environment, banks can In improve their profitability by increasing income from non-core activities such as service fees, commissions, and transaction costs. Studies such as Yüksel et al. (2018) established that there is positive association between bank profits and non-interest income. Besides, the profitability of banks in Zimbabwe seems to be spurred by non-interest income given the significant contribution of non-interest income to total bank profits over the past years. Therefore, this study anticipates a positive and significant impact of non-interest income on bank profitability in Zimbabwe.

Liquidity (*liq*). According to the risk-return trade-off theory, there is a negative relationship between bank profitability and liquidity. This theory assumes that liquid assets have lower returns than illiquid assets, as they entail lower risks (Bordeleau & Graham, 2010). Therefore, the study hypothesizes that banks with higher liquidity (low loan-to-deposit ratio) are less profitable than banks with lower liquidity. However, some studies have challenged this hypothesis and found a positive relationship between liquidity and profitability (Tran et al., 2016; Goddard et al., 2013). They argued that liquid assets can help banks cope with unexpected shocks and reduce the cost of external funding (Dietrich & Wanzenried, 2011; Berger & Bouwman, 2009). Following Satria et al. (2016) and Bordeleau and Graham (2010), the study measures bank liquidity by the loan-to-deposit ratio, which indicates how much of a bank's deposits are used to make loans. A higher loan-to-deposit ratio means that the bank has more loans than deposits, implying lower liquidity, and vice versa.

Operational efficiency (cost-to-income ratio, cost_eff). Operational efficiency measured by the cost-to-income ratio is another variable the study predicts have a significant impact to on the profitability of banks. The cost-to-income ratio indicates management's efficiency in controlling costs. A high ratio demonstrates that management is inefficient in controlling costs which adversely Lakew, profitability (Rao & affects 2012). Accordingly, the study expects an inverse association between operational efficiency and bank profit. Previous studies have also confirmed the negative effect of operational efficiency on bank profitability in different contexts (Flamini et al., 2009; Sufian & Habibullah, 2009; Athanasoglou et al., 2008).

Bank capital (cap). Capital is a vital source of funding for banks, as it enables them to absorb losses, maintain solvency, and comply with regulatory requirements (Li & Feng, 2016; Ahlswede & Schildbach, 2012). Moreover, high capital is associated with greater stability and confidence, which can attract more deposits and reduce the cost of funding (Dietrich & Wanzenried, 2011; Berger et al., 2013). Therefore, one would expect banks with high levels of equity capital to have more funds to lend out, which should result in increased profitability. Accordingly, the study hypothesizes that capital positively affects bank profitability. Bank capital is measured by the tier 1 ratio, which indicates the ratio of a bank's core equity capital to its total risk-weighted assets.

Credit risk (loan loss reserves, llr). Credit risk is the risk of loss arising from the failure of borrowers to repay their loans or meet their contractual obligations. Credit risk is one of the major sources of income and risk for banks, as lending is their core business activity (Dang & Nguyen, 2022; DeYoung & Rice, 2004). However, if a bank fails to properly



manage its credit risk, loan defaults erode its profits and capital (Dang & Nguyen, 2022; Subhanij, 2010). Thus, the study claims that credit risk negatively affects the performance of banks in Zimbabwe. Credit risk is proxied by the loan loss reserve ratio, which measures the ratio of loan loss provisions to total loans. Recent studies have also examined the impact of credit risk on bank profitability in different regions and contexts, such as Africa (Opoku-Mensah et al., 2019), Asia (Farooq et al., 2021), Europe (Kosmidou, 2008), and Islamic banking (Al-Harbi et al., 2019).

Economic activity (*GDP*). Economic activity, measured by real GDP, reflects the level of output and income in an economy. Economic activity has a cyclical relationship with bank profitability, as it affects both the demand and supply of credit, as well as the quality of borrowers (Satria et al., 2016; Albertazzi & Gambacorta, 2009). During economic downturns, credit quality wanes leading to higher loan defaults (Yarovaya et al., 2020), thereby reducing bank profitability. In times of economic booms, demand for loans generally rises coupled with improving credit quality, leading to improved bank profitability (Albertazzi & Gambacorta, 2009). As such, a positive relationship between GDP and bank profit is anticipated. This study uses the GDP growth rate to proxy economic activity. Some recent studies have also explored the role of economic activity in determining bank profitability in various countries and regions, such as China (Zhang et al., 2022), India (Almaqtari et al., 2019), Latin America (Jara-Bertin et al., 2014), and Sub-Saharan Africa (Flamini et al., 2009).

Inflation (inf). Inflation is the general increase in the prices of goods and services over time. The relationship between inflation and bank profit is ambiguous, as it depends on the banks' ability to forecast inflation developments and adjust their interest rates accordingly (Yüksel et al., 2018). If banks can accurately predict inflation trends, they can increase their revenues faster than their costs and reap higher profits. However, if banks fail to anticipate inflation changes or face regulatory constraints on interest rate settings, they may suffer from lower margins and reduced profitability. Hence, this study expects either a positive or negative relationship between inflation and bank profitability. The annual inflation rate is used in this study. Some recent studies have also investigated the effect of inflation on bank profitability in different settings, such as Turkey (Yüksel et al., 2018), Pakistan (Farooq et al., 2021), and Nigeria (Ogunmuyiwa & Ekone, 2010).

Market structure (*c3*). The study also examines the effect of market structure on bank profitability in Zimbabwe. Based on the structure-conductperformance (SCP) hypothesis, market structure influences the competitive behaviour of banks, which in turn affects their profitability. According to this hypothesis, a more concentrated banking market (i.e., a market with fewer and larger banks) leads to higher profitability, as banks can collude with each other to charge higher interest rates and fees, and reduce their operating costs (Kosmidou, 2008). Therefore, the study predicts a positive nexus between market share and bank profitability and measures market structure using the *c3* ratio, which is the market share of the three largest banks by asset size.

Bank stability (zscore). The study explores the relationship between bank stability, measured by the *zscore*, and bank profitability in Zimbabwe. Bank stability can impact bank profitability through two channels: the cost of funding and risk-taking activities. A more stable bank can attract greater deposits and reduce its funding costs, as depositors perceive it as less likely to default or encounter liquidity problems. This lower cost of funding can in turn increase the net interest margin and overall profitability of the bank. For instance, Nguyen and Le (2022) and Nisar et al. (2018) found that bank stability *zscore* has a positive and significant effect on bank profitability in Asian economies. Thus, the study hypothesizes that financial stability has a positive effect on bank profitability.

3.4. Empirical specification

The empirical model (Eq. (2)) incorporates several key elements to enhance its accuracy and robustness. Firstly, it accounts for unobserved bank-specific fixed effects, which capture factors such as management quality and board effectiveness that can significantly influence banks' performance. By including these fixed effects, the model ensures that the estimated coefficients are not biased. Additionally, the model addresses the issue of persistence in bank performance, which can arise from market structure imperfections like cartels and monopolies. These imperfections can lead to sustained differences in bank performance over Acknowledging and incorporating time. this persistence into the model, it provides a more comprehensive understanding of the determinants of bank performance.

Moreover, the model recognizes the possible presence of endogeneity, wherein the explanatory variables may be correlated with the idiosyncratic error term. To address this concern, the study employs the GMM estimator. This estimator effectively controls for endogeneity and provides consistent estimates. Within the GMM framework, prominent estimators employed: two are the difference GMM (Arellano & Bond, 1991) and the system GMM (Arellano & Bover, 1995; Blundell & Bond, 1998). While both estimators have their merits, the system GMM is preferred in this study due to its ability to eliminate fixed effects using forward orthogonal deviation transformations. This transformation ensures consistent estimates and enhances the model's performance compared to the difference GMM (Blundell & Bond, 1998). The study uses the xtabond2 command in Stata to implement the system GMM estimator. One of the main advantages of xtabond2 is that it can control for instrument proliferation, which is a common problem in dynamic panel data models (Roodman, 2006). Instrument proliferation occurs when the GMM estimation uses too many instruments compared the number to of observations or groups. This can cause problems such as overfitting, weak identification, and biased inference. xtabond2 has several options to reduce the number of instruments, such as collapsing the instrument matrix, limiting the lag depth, or using principal components (Roodman, 2006).



4. RESEARCH FINDINGS

4.1. Descriptive statistics

Table 1 presents the descriptive statistics of the data used in the study. The table shows both the raw and the log-transformed values of the variables to compare their distributions before and after the log transformation. However, the analysis is based on the raw data, as it reflects the original scale and variation of the variables.

Non-transformed data							
Variable	Obs.	Mean	Std. dev	Minimum	Maximum		
nim	113	11.25	5.10	3.61	31.25		
llr	102	4.76	3.28	0.44	20.07		
сар	83	20.71	10.75	5	60		
nii	113	12.33	9.52	2.65	47.29		
cost_eff	113	70.45	27.95	19.34	253.33		
liq	113	38.63	18.62	5.85	81.02		
с3	102	59.54	5.49	52.71	81.83		
inf	73	3.75	5.63	-6.25	16.67		
GDP	79	0.51	4.27	-6.25	4.82		
zscore	104	5.91	1.07	5.01	8.08		
Log-transformed data							
Variable	Obs.	Mean	Std. dev	Minimum	Maximum		
nim	113	2.33	0.44	1.28	3.44		
lnllr	102	1.37	0.63	-0.82	3.00		
Incap	83	2.91	0.51	1.61	4.09		
lnnii	113	2.25	0.73	0.98	3.86		
lncost_eff	113	4.20	0.34	2.96	5.53		
Inliq	113	3.51	0.60	1.77	4.39		
lnc3	102	4.08	0.09	3.96	4.40		
lninf	73	1.07	0.91	-0.28	2.81		
InGDP	79	1.16	0.94	-0.28	2.98		
Inzscore	104	1.76	0.17	1.61	2.09		

Table 1. Descriptive statistics

Source: Authors' elaboration.

The results Table 1 above show in the descriptive statistics of the net interest margin (nim) for the sampled banks over the period 2011 to 2020. The nim is a measure of bank profitability that indicates the difference between the interest income and the interest expense as a percentage of total assets. The mean value of nim for the sampled banks is 11.25%, which suggests that the banks have a high-interest margin on average. However, the standard deviation of *nim* is 5.10%, which implies that there is a large variation in profitability across banks. The minimum value of nim is 3.61%, while the maximum value is 31.25%, indicating that some banks have very low or very high profitability compared to others. The range of nim values is 27.64%, which reflects the wide dispersion of profitability in the sample.

The loan loss reserves (*llr*) to total loans ratio, which averaged 4.76% among the banks included in the study, indicates that banks in Zimbabwe set

aside reserves of approximately 5% to account for potential loan defaults. The small standard deviation of 3.28% suggests that there is little variation in the amount of loan reserves held by the sampled banks.

Based on an average tier 1 capital (*cap*) ratio of 20.71%, it can be concluded that banks in Zimbabwe are adequately capitalized according to the minimum tier 1 ratio of 8% recommended by the Basel II^2 framework. However, the presence of minimum and maximum values of 5% and 60% respectively is concerning, as it reveals that some banks in the sample have low capital while others are highly capitalized.

On average, banks in the study generated about 12% of their total revenue from non-interest income (*nii*) activities. This implies that 12% of a bank's income is derived from sources other than interest, while the majority, 88%, comes from interest sources. This heavy reliance on interest income indicates a low diversification of revenue sources for the banks, which could expose them to interest rate risk. Such risk arises from changes in interest rates that can impact the bank's profitability and overall value.

Examining Table 1, the loan-to-deposit ratio (*liq*) averaged 39% for the sampled banks during the study period. This ratio is concerning, suggesting constrained credit extension by banks possibly due to the volatile macroeconomic environment in Zimbabwe. Additionally, the minimum ratio of 6% implies liquidity hoarding by banks in the country.

With an average cost-to-income ratio (*cost_eff*) of 70%, it is evident that operational costs for banks in Zimbabwe are very high. This can be attributed to the predominantly high-cost structure of the country's economy, influenced by factors such as high energy costs, inflation, and high taxes.

From an economic perspective, Zimbabwe has experienced modest growth during the sampling window, with an average growth rate of 0.51%. This subdued growth can be attributed to various factors, including political instability, currency volatility, and high inflation.

4.2. Correlation matrix

Table 2 shows the pairwise correlation coefficients between the variables used in the study. The correlation matrix is widely used to detect multicollinearity, which is a problem that occurs when two or more variables are highly correlated with each other, causing instability and bias in the regression estimates. Besides this, it can help to identify the potential relationships between the variables.

² Banks in Zimbabwe have not yet fully adopted the Basel III standards, which are a set of international regulations for banking supervision and risk management. They are still operating under the Basel II framework, which is the previous version of the standards. However, they are in the process of transitioning to Basel III compliance, which is expected to enhance their resilience and stability.

	lnnim	Inllr	Incap	lnnii	lncost_eff	Inliq	Inc3	lninf	InGDP	Inzscore
Innim	1.00									
Inllr	-0.15	1.00								
Incap	0.07	0.03	1.00							
Innii	0.51*	-0.16	0.13	1.00						
Incost_eff	0.02	0.09	0.10	0.24*	1.00					
Inliq	-0.13	-0.09	-0.63*	-0.52*	-0.20*	1.00				
lnc3	0.17	-0.07	-0.35*	0.04	-0.14	0.39*	1.00			
lninf	0.13	-0.18	-0.01	0.24*	-0.08	-0.28*	-0.03	1.00		
InGDP	0.02	-0.08	-0.34*	0.26*	0.002	0.03	0.27*	0.36*	1.00	
Inzscore	0.58*	-0.22*	0.17	0.76*	0.19	-0.51*	0.03	0.39*	0.10	1.00

Table 2. Correlation matrix

Note: * *indicate statistical significance at the 5% level.*

Source: Authors' elaboration.

The results in Table 2 show that there are no variables with a correlation above 0.70, which is a common threshold for detecting multicollinearity. Therefore, it can be concluded that the research data free from multicollinearity is and that the explanatory variables are sufficiently independent from each other. Besides detecting multicollinearity, the correlation matrix also provides some insights into possible the relationships among the variables. Some of the interesting and significant correlations are as follows:

The variable *lnnim* has a positive and significant correlation with *lnnii* and *lnzscore*, which means that banks with higher net interest margins tend to have higher non-interest income and higher *zscore*. This is consistent with the literature that suggests that *lnnim* is a proxy for bank profitability and stability.

The variable *lncap* has a negative and significant correlation with *lnliq*, which means that banks with higher capital adequacy ratios tend to have lower loan-to-deposit ratios. This may indicate that banks with higher capital have lower liquidity needs or preferences, or that they face lower demand for loans or a higher supply of deposits.

Interestingly, the variable *lnnii* has a positive and significant correlation with *lncost_eff*, meaning that banks with higher non-interest income tend to have a higher cost-efficiency ratio. This may suggest that banks with higher non-interest income have higher operating costs or lower operating income, or that they invest more in technology and innovation to generate non-interest income.

Bank concentration, proxied by *lnc3*, has a positive and significant correlation with *lnGDP*, which means that banks in more concentrated markets are characterized with higher gross domestic product growth rate. This may imply that banks in more concentrated markets face lower competition or higher regulation, which may affect their liquidity management or intermediation function. The variable *lnzscore* has a positive and significant correlation with nim, lnnii, and lninf, which means that banks with higher *zscore* tend to have higher net interest margin, higher non-interest income, higher cost efficiency ratio, and higher inflation. This may suggest that *lnzscore* is a proxy for bank stability and performance, as higher zscore may reflect higher profitability, solvency, diversification, resilience, and competitiveness of banks.

Turning to the macroeconomic fundamentals, *lninf* has a positive and significant correlation with InGDP, and Inzscore, which means that banks in countries with higher inflation rates tend to have higher gross domestic product growth rate, and higher *zscore*. On the other hand, the variable *lnGDP* has a positive and significant correlation with *lnnii*, *lnc3*, and *lninf*, which means that banks in countries with higher gross domestic product growth rate tend to have higher non-interest income, higher concentration ratio, and higher inflation rate. This may suggest that *lnGDP* is a proxy for economic growth and financial performance, as higher GDP growth may indicate higher income and consumption levels, as well as higher financial inclusion and stability.

The correlation matrix provided some preliminary evidence of the relationships between the dependent variable (bank profitability) and the explanatory variables (bank-specific, industry, and macroeconomic factors). However, it does not establish any causal effects or control for other confounding factors. To address these issues, the study uses a panel data regression analysis to estimate the effects of the explanatory variables on bank profitability in Zimbabwe. The results and discussion of the regression analysis are presented in the next section.

4.3. Empirical findings

4.3.1. Main results

This section presents the results of estimating Eq. (2) with system GMM, which examines the determinants and dynamics of bank profitability. The study uses the Sargan test to check for instrument validity. The Sargan test statistic has a p-value of 0.13, which does not reject the null hypothesis of no overidentification at the 5% significance level. This indicates that the instruments used in the model are valid and exogenous. The study tests for the absence of second-order autocorrelation in first-differenced errors using the Arellano and Bond (1991) test (AR (2) test). The AR (2) test statistic has a p-value of 0.13, which does not reject the null hypothesis of no second-order autocorrelation at the 5% significance level. This indicates that the model satisfies the assumptions for using system GMM. The results are shown in Table 3 below.



Variable	Coefficient	Std. error		
lnnim(1)	0.6212***	0.1953		
lnllr	-0.1153*	0.0607		
lncap	0.1435**	0.0655		
lnnii	0.1512***	0.0459		
lncost_eff	0.1248***	0.0459		
lnliq	0.3503***	0.1046		
lnc3	-0.7032**	0.2810		
lninf	0.0469	0.0369		
InGDP	0.0279	0.0294		
lnzscore	0.8000*	0.4379		
Diagnostics				
AR(2)	0.13			
Sargan	0.13			

Table 3. Empirical findings

Note: ***; **; * indicate statistical significance at the 10%, 5%, and 1% levels.

Source: Authors' elaboration.

The analysis shows that the net interest margin (*Innim*), which is a measure of bank profitability, has a significant positive relationship with its lagged value (*lnnim*(1)). This means that banks in Zimbabwe tend to maintain their profitability levels over time and adjust slowly to changes in market conditions or bank-specific factors. The coefficient of the lagged *nim* is 0.6212, indicating that a 1% increase in the previous year's nim leads to a 0.6212% increase in the current year's nim. The speed of adjustment, which measures how fast banks converge to their desired or optimal profitability levels, is 0.38. This implies that banks in Zimbabwe are relatively slow in adjusting their profitability and eliminate only 38% of the gap between their actual and desired nim in a year. This could be due to various reasons, such as high adjustment costs, market frictions, regulatory constraints, or managerial inertia. This finding is important because it reveals how banks in Zimbabwe manage their profitability and respond to changes in the banking sector.

The results show that the logarithm of non-interest income (Innii) has a positive and statistically significant impact on bank profitability in Zimbabwe. In particular, the coefficient of Innii is estimated to be 0.1512, with a p-value of 0.0000, indicating that this relationship is highly significant. These results indicate that for every 1% increase in the ratio of non-interest income to total income, there is a corresponding increase in net interest margin (nim) of 15% points. This evidence supports the notion that diversifying income sources and reducing reliance on interest income can be beneficial for banks in Zimbabwe. By expanding their revenue streams beyond traditional interestbased activities, banks can not only enhance profitability but also improve their overall stability. This rationale is intuitive since non-interest income is known to be less responsive to business cycles. The findings of this study align with the perspective put forth by Albertazzi and Gambacorta (2009). which emphasizes the importance of non-interest income in fostering bank stability. Their argument is grounded on the premise that incorporating diverse income streams can help buffer banks from the volatilities associated with interest-based activities.

The impact of liquidity on bank profitability in Zimbabwe can be seen through the natural logarithm of this ratio (Inlig). The coefficient of Inlig is noted as 0.3503, which holds a significant and positive influence as indicated by a p-value of 0.0000. Consequently, a 1% increase in loan-to-deposit ratio equates to a 35% point rise in nim. This observation suggests that banks in Zimbabwe experience advantages from having a higher proportion of loans compared to deposits, resulting in increased interest income and decreased interest expense. This suggests that banks in Zimbabwe are able to charge higher interest rates on their loans than they pay on their deposits, reflecting their market power and pricing strategy. A similar finding by Isayas (2022) in Ethiopia supports these results, further affirming the positive and significant relationship between liquidity and bank profitability.

Contrary to popular belief, the variable cost-to-income ratio (*lncost_eff*) has a notable positive impact on bank profitability in Zimbabwe. This is supported by a statistically significant point estimate of 0.1248. These findings challenge existing literature, such as the studies conducted by Al-Tarawneh et al. (2017) and Athanasoglou et al. (2008), which suggest that a higher cost-to-income ratio hampers bank performance. One possible this unexpected explanation for result is the presence of market power among Zimbabwean banks. This enables them to impose high interest rates on loans while offering low-interest rates on deposits. This aligns with Abel et al. (2018), who demonstrate that banks in Zimbabwe possess considerable market power, as measured by the Lerner index, and how it affects their profitability over time.

The coefficient of the capital adequacy ratio (*lncap*), measured by the tier 1 ratio, is found to be statistically significant at 0.1435. This means that a 1% change in the capital ratio results in a 14% point increase in bank profits. These results suggest a positive relationship between capital and bank profitability, which is consistent with the findings of García-Herrero et al. (2009). García-Herrero et al. (2009) argue that banks with higher capital ratios are able to reduce their funding costs as they face a lower risk of bankruptcy. Thanh et al. (2022) also support this notion, stating that higher equity gives banks an advantage in raising capital, improves their risk tolerance, and enhances customer confidence, ultimately boosting profitability. Jadah et al. (2020) provide similar evidence for Iraqi banks.

The empirical results presented in Table 3 provide support for the hypothesis that credit risk has a significant negative impact on bank performance. The coefficient of -0.12 on the loan loss reserves ratio (*lnllr*) indicates that a 1% increase in *llr* leads to a decline of approximately 12% in net interest margin (*lnnim*), a measure of bank profitability. This finding aligns with previous research, including studies by Siddique et al. (2021) and Garcia-Herrero (2009), which have identified a negative relationship between credit risk and bank performance. The negative effect of credit risk on bank performance can be attributed to the fact that higher *llr* reflects a greater number of nonperforming loans (NPLs), reducing income from increasing provision loans and expenses. Consequently, it is essential for banks to implement effective credit risk management practices to minimize NPLs and enhance their overall performance.

The study shows that the *c3* ratio coefficient demonstrates a negative association with *nim* that is

NTERPRESS VIRTUS

statistically significant. The coefficient value is found to be -0.7032, with a p-value less than 0.05. This implies that market structure plays a significant role in negatively impacting bank profitability in the country. These findings align with those of Bolarinwa et al. (2019), who also found that elevated levels of market concentration, as measured by the Herfindahl-Hirschman index (HHI), have a detrimental effect on bank profitability. Considering these results, it can be inferred that Zimbabwean banks encounter substantial challenges in the form of high operational costs, credit risks, and regulatory burdens. These challenges seem to outweigh the potential benefits that could arise from market concentration. Consequently, the banking sector in Zimbabwe must focus on enhancing its efficiency, fostering innovation, and strengthening its risk management practices. These areas of improvement are crucial for enhancing profitability and fortifying the sector's resilience in the face of such demanding market conditions.

The statistically significant coefficient of 0.80 on the variable zscore strongly supports the hypothesis that financial stability promotes bank stability. This finding is intuitive and aligns with the belief that greater stability leads to increased bank profitability. Consequently, it is evident that enhancing the resilience and efficiency of the financial system can contribute to the stability and performance of banks. This conclusion is consistent with the findings of Nguyen and Le (2022) and Nisar et al. (2018), who also observed a positive and significant impact of bank stability (measured by the zscore) on bank profitability in Asian economies. As a result, our results suggest that policymakers and regulators should implement measures to foster financial stability, such as strengthening macroprudential supervision, improving crisis management and resolution frameworks, and promoting international cooperation.

Turning to macroeconomic fundamentals, the empirical analysis conducted for this study yielded findings that are inconsistent with initial expectations. The first variable examined is economic activity, as measured by gross domestic product (GDP) growth. It was expected that an increase in GDP growth would be associated with higher bank profitability. However, the coefficient for this variable was found to be statistically insignificant at conventional levels, indicating that there is no strong evidence to support the notion that bank profits in Zimbabwe are cyclical in nature. Previous studies conducted by Ozili (2015), Boitan (2015), and Albertazzi and Gambacorta (2009) have demonstrated a positive relationship between business cycles and bank profitability. However, the current analysis did not find similar evidence, suggesting that business cycles may not have a significant effect on bank profitability in Zimbabwe. This finding is particularly noteworthy given the challenging economic environment in the country, highlighting the ability of Zimbabwean banks to implement effective strategies that enable them to withstand tough business conditions.

Moving on to the analysis of inflation (*lninf*), it was hypothesized that inflation would have a negative impact on bank profitability in Zimbabwe. However, the coefficient for inflation in the regression model was found to be 0.0469, which is statistically insignificant at conventional levels. This suggests that there is no compelling evidence to support the claim that inflation negatively affects the profitability of banks in Zimbabwe. The findings align with Bolarinwa et al. (2019), who also reported a positive but insignificant effect of inflation on profitability in Nigeria. One possible bank explanation for these findings is that banks in Zimbabwe have been adept at predicting and incorporating inflation forecasts into their interest rate structures. It is plausible that policy inconsistencies in the country have enabled banks to successfully adapt and build inflation forecasts into their operations. This ability to navigate inflationary pressures may contribute to the resilience of Zimbabwean banks and their ability to maintain profitability even in the face of economic challenges. In summary, the empirical analysis conducted in this study did not find strong evidence to support the hypotheses that bank profits in Zimbabwe are cyclical or that inflation negatively affects bank profitability. These findings suggest that banks have been successful in Zimbabwean implementing sound strategies that enable them to withstand tough business conditions and effectively manage inflationary pressures.

4.3.2. Moderation analysis

This study extends the previous literature on this topic by examining the moderating effect of bank stability and fintech on the determinants of bank profitability in Zimbabwe. The results are presented in Table 4 below. Column 2 shows the results of the regression model that includes the interaction terms between bank stability and regulatory capital, while column 3 shows the results of the regression model that includes the interaction terms between fintech and non-interest income.

Variable	Model 1	Model 2		
	(2)	(3)		
nim(1)	0.7249***	0.6645***		
lnllr	-0.0773	0.00004		
lncap	0.4480***	0.2379*		
lnnii	0.1900***	0.1461*		
lncost_eff	0.0291	0.0483		
lnliq	0.2705**	0.2131*		
lnc3	-0.9249***	-0.2759*		
lninf	0.0441	0.0416		
InGDP	0.0433	0.0403*		
zscore	1.0821**	-		
reg_zscore	-0.0020**	-		
fintech	-	-0.0443		
fintech_nii	-	-0.0001		
Diagnostics				
AR(2)	0.213	0.082		
Sargan	0.142	0.136		
Note: ***; **; * indicate statistical significance at the 10%, 5%, and				

Table 4. Moderation analysis results

1% levels. Source: Authors' elaboration.

The variable *reg_zscore* is the interaction term between the regulatory capital and the *zscore* variables. The coefficient of *reg_zscore* is -0.002, which is negative and significant at the 5% level. This means that there is a weak negative and significant moderating effect of regulatory capital on the relationship between bank stability and bank profitability. In other words, higher regulatory capital weakens the positive effect of bank stability on bank profitability. This result may suggest that higher regulatory capital reduces the risk-taking incentives or opportunities of banks, which may lower their profitability potential. This analysis is consistent with the literature. For example, Agénor and da Silva (2021) developed a model of endogenous growth with banking, limited liability, and government guarantees. They showed that higher capital requirements increase the skin-in-thegame effect, which reduces the default risk and the incentive for banks to extend risky loans. They also found that the optimal capital adequacy ratio may be too high in practice and may require other regulatory measures to prevent distortions in financial markets.

The variable *fintech_nii* is the interaction term between the fintech and the non-interest income. The coefficient of *fintech_nii* is -0.0001, but it is statistically insignificant at the conventional levels. This means that the study could not find evidence for the moderating effect of fintech on the relationship between non-interest income and bank profitability. This result may imply that fintech does not have a significant impact on the market share or pricing power of banks in non-interest income activities, or that it has offsetting effects on their costs and revenues (Philippon, 2016).

4.3.3. Robustness analysis

To ensure the validity and reliability of our empirical findings, we conducted a number of robustness tests that are presented in Table 5 below. For easy comparison, the baseline results are displayed in column 2. First, we added money supply (ms) as an additional variable in column 3. Money supply has been a key driver of inflation in Zimbabwe over the past decades (Maune et al., 2020), so we wanted to control its effect on bank performance. Second, we replaced the loan loss reserve ratio with impaired loans (il) as a measure of asset quality in column 4. Impaired loans reflect the actual quality of the loan portfolio better than the loan loss reserve ratio, which may be subject to managerial discretion. Third, we used *c5* instead of *c3* as a measure of bank concentration in column 5. The variable *c5* captures the market share of the top five banks more accurately than c3, which only considers the top three banks.

 Table 5. Robustness test results

Variable	Model 1	Model 2	Model 3	Model 4		
	(2)	(3)	(4)	(5)		
nim(1)	0.6212***	0.6675***	0.7162***	0.6277***		
Inllr	-0.1153*	-0.1113*	-	-0.1156*		
Incap	0.1435**	0.1258*	0.1568**	0.1392**		
Innii	0.1512***	0.1535***	0.1597***	0.1494**		
lncost_eff	0.1248***	0.1233**	0.1050***	0.1260***		
Inliq	0.3503***	0.3351***	0.3820***	0.3433***		
lnc3	-0.7032**	-0.7066***	-0.7392***	-		
lnc5	-	-	-	-0.6595**		
lninf	0.0469	0.0496	0.0401	0.0473		
InGDP	0.0279	0.0302	0.0460*	0.0267		
Inzscore	0.8000*	0.7852*	0.6617	0.8029*		
ms	-	0.0189	-	-		
il	-	-	-0.0677*	-		
Diagnostics						
AR(2)	0.13	0.117	0.129	0.132		
Sargan	0.13	0.131	0.167	0.119		

Note: ***; **; * indicate statistical significance at the 10%, 5%, and 1% levels. Source: Authors' elaboration.

In this section, we present the results of our robustness tests and discuss their implications for our main findings. First, we included money supply (ms) as an additional variable in Model 2 (column 3) because it has been a key driver of inflation in Zimbabwe over the past decades, which may affect bank performance. However, these results suggest that money supply has no significant effect on bank profitability in Zimbabwe. One possible explanation of this evidence is that money supply is endogenous to bank profitability (Sieroń, 2019), meaning that it is determined by the demand and supply of money in the economy, which is influenced by bank behaviour. Second, we used impaired loans (il) instead of loan loss reserves (llr) as a measure of asset quality in Model 3 (column 4). We used impaired loans as an alternative measure of asset quality because it reflects the actual quality of the loan portfolio better than the loan loss reserves, which may be subject to managerial discretion. The results show impaired loans have a negative and significant effect on bank profitability, reaffirming that poor credit risk

management is detrimental to bank profitability. Third, we used c5 instead of c3 as a measure of bank concentration in Model 4 (column 5). We used c5 as a different measure of bank concentration because it captures the market share of the top five banks more accurately than c3, which only considers the top three banks. The results show a significant negative effect of bank concentration on bank profitability in Zimbabwe. This supports earlier findings that high concentration dampens bank profitability in Zimbabwe.

5. CONCLUSION

This study examines the factors that affect the profitability of banks in Zimbabwe, which have shown remarkable performance despite the challenging economic conditions. The study uses a panel data analysis of 11 banks over the period 2011–2020 and employs the net interest margin as a measure of bank profitability. The study considers various bank-specific, industry, and macroeconomic

variables as potential determinants of bank profitability. The bank-specific variables that are considered in this study are non-interest income, liquidity, cost-to-income ratio, tier 1 capital, bank stability, and asset quality. To measure the degree of competition in the banking sector, the study uses bank concentration as a proxy for market structure. The macroeconomic variables include GDP and inflation. The study finds that non-interest income, liquidity, cost-to-income ratio, tier 1 capital, and bank stability have a positive and significant impact on bank profitability, while asset quality and bank concentration have a negative and significant impact on bank profitability. The study also reports that GDP and inflation do not have a significant effect on bank profitability. The study concludes that bank profitability in Zimbabwe is mainly driven by internal factors that are under the control of bank managers and regulators, rather than external factors that are beyond their influence. The study provides several policy implications and recommendations based on the findings.

non-interest evidence that The income positively affects bank profitability, suggests that banks can boost their revenues by diversifying their income sources beyond interest-based activities. Policymakers can support this strategy by facilitating the development and promotion of innovative banking products and services that cater to the needs of customers. Another positive determinant of bank profitability is liquidity, which reflects the ability of banks to meet their short-term obligations and fund their lending activities. To enhance their profitability, banks need to maintain a healthy level of liquidity that balances the demand for loans and the availability of deposits. Policymakers and regulators can assist this process by providing guidelines and mechanisms that ensure adequate liquidity management and prevent liquidity crises.

The study also finds a positive relationship between the cost-to-income ratio and bank profitability, implying that efficiency is a key factor in improving bank performance. Banks can achieve higher efficiency by reducing their operating costs and increasing their productivity. This can be done by streamlining operations, improving processes, and leveraging technology to optimize their cost structure and enhance their profitability. Another factor that positively influences bank profitability is tier 1 capital, which measures the core capital of banks that is used to absorb losses and protect depositors. The study suggests that banks with higher capital adequacy are more profitable than those with lower capital adequacy. Therefore, policymakers should recommend and enforce regulations that ensure banks comply with the minimum capital requirements and maintain sufficient capital buffers to mitigate risks and absorb losses.

The positive effect of bank stability on bank profitability underscores the importance of maintaining a stable and resilient banking system. Policymakers should implement measures to strengthen regulatory oversight, monitor banks' risk management practices, and promote financial stability to support bank profitability.

On the other hand, the study identifies two negative determinants of bank profitability: asset

quality and bank concentration. The negative effect of loan loss reserves on bank profitability highlights the need for a robust credit risk management framework. Banks should adopt prudent lending practices, conduct thorough credit assessments, and establish adequate loan loss reserves to mitigate potential losses. Policymakers should reinforce regulations and supervision to ensure banks adhere to sound credit risk management practices. The study reveals that banks operating in a more concentrated market are less profitable than those operating in a more competitive market. To increase their profitability, banks need to expand their market share and customer base by offering better products and services at competitive prices. outcome Policymakers can foster this by encouraging market competition, promoting fintech, and discouraging excessive market concentration that may harm overall bank profitability.

The study also reports that macroeconomic fundamentals, such as GDP and inflation, do not have a significant impact on bank profitability in Zimbabwe. This implies that macroeconomic factors have a limited direct influence on banks' financial performance in the country. However, policymakers should still monitor macroeconomic conditions as they indirectly affect the overall business environment and determine the demand for banking services.

Finally, the analysis reveals that the impact of regulatory quality on the relationship between bank stability and profitability is weakly negative and implies significant. This that enhancing the regulatory environment may not necessarily improve the performance and resilience of banks in Zimbabwe. Policymakers and regulators need to carefully consider the trade-offs and costs associated with imposing stricter regulations and supervision on the banking sector. It is essential to strike a balance between prudential norms, transparency requirements, and consumer protection measures, while also considering the impact on profitability incentives, risk-taking opportunities, and innovation capabilities of banks. Furthermore, the analysis indicates that fintech has insignificant moderating effect an on the relationship between non-interest income and bank profitability. This suggests that fintech may not significantly impact the competitiveness and performance of banks in Zimbabwe in non-interest income activities. Bank managers and practitioners should closely monitor trends and developments in fintech within the country and the region. They need to assess the potential opportunities and challenges that fintech presents for their business models, products, and services. Additionally, exploring the possibilities of collaborating or partnering with fintech providers can help enhance customer experience, loyalty, and satisfaction.

One of the main limitations of this study is the data availability and quality. Due to the economic and political instability in Zimbabwe, reliable and consistent data on the banking sector and the macroeconomic environment are scarce and difficult to obtain. This limits the sample size and the time span of the analysis, as well as the choice of variables and methods. For example, we could not include some important factors that may affect bank profitability, such as competition, efficiency, and



innovation. We also could not perform some desired robustness tests or sensitivity analyses that would enhance the validity and reliability of our results. Therefore, our findings should be interpreted with caution and subject to further verification with more comprehensive and updated data.

In recent years, the Sub-Saharan African region has experienced tremendous growth in fintech activities, particularly in the realm of mobile money. As of now, the region is considered the global leader in mobile money initiatives. The emergence and evolution of fintech in this region has disrupted traditional banking systems, leading to several innovative financial services. However, despite this significant progress, the impact of fintech developments on the profitability of banks in Sub-Saharan Africa remains uncertain. The rapid evolution of fintech, combined with its dynamic nature, presents opportunities, challenges, and, inevitably, risks for traditional banking institutions. Therefore, a comprehensive investigation is warranted to understand the potential effects of fintech on the financial landscape and profitability specifically within Sub-Saharan African banking sectors.

REFERENCES

- Abbas, F., Iqbal, S., & Aziz, B. (2019). The impact of bank capital, bank liquidity and credit risk on profitability 1. in post-crisis period: A comparative study of US and Asia. Cogent Economics & Finance, 7(1), Article 1605683. https://doi.org/10.1080/23322039.2019.1605683
- Agénor, P. R., & da Silva, L. A. P. (2021). Capital requirements, risk-taking and welfare in a growing economy. 2.
- *Journal of Regulatory Economics*, 60(2-3), 167-192. https://doi.org/10.1007/s11149-021-09438-z Ahlswede, S., & Schildbach, J. (2012). *Poised for a comeback: Bank deposits* (Deutsche Bank Research). https://www.dbresearch.com/PROD/RPS_ENPROD/PROD000000000454753/Poised_for_a_comeback%3A_Bank 3. _deposits.PDF
- Albertazzi, U., & Gambacorta, L. (2009). Bank profitability and the business cycle. Journal of Financial Stability, 4. 5(4), 393-409. https://doi.org/10.1016/j.jfs.2008.10.002
- 5. Al-Harbi, A. (2019). The determinants of conventional banks profitability in developing and underdeveloped OIC countries. *Journal of Economics* https://doi.org/10.1108/JEFAS-05-2018-0043 Economics, Finance and Administrative Science. 24(47).4 - 28.
- Al-Jafari, M. K., & Alchami, M. (2014). Determinants of bank profitability: Evidence from Syria. Journal of 6. Applied Finance and Banking, 4(1), 17-45.
- Ally, Z. (2014). Determinants of banks' profitability in a developing economy: Empirical evidence from 7. Tanzania. European Journal of Business and Management, 6(31), 363-375. http://surl.li/mutbg
- Almaqtari, F. A., Al-Homaidi, E. A., Tabash, M. I., & Farhan, N. H. (2019). The determinants of profitability of 8. Indian commercial banks: A panel data approach. International Journal of Finance & Economics, 24(1), 168-185. https://doi.org/10.1002/ijfe.1655
- Al-Tarawneh, A., Abu Khalaf, B. K., & Al Assaf, G. (2017). Noninterest income and financial performance at Jordanian banks. *International Journal of Financial Research*, 8(1), 166–171. https://doi.org/10.5430 9. /ijfr.v8n1p166
- 10. Arellano, M., & Bond, S. (1991). Some tests of specification for panel data: Monte Carlo evidence and an application to employment of Economic equations. The Review Studies 58(2). 277-297. https://doi.org/10.2307/2297968
- 11. Arellano, M., & Bover, O. (1995). Another look at the instrumental variable estimation of error-components models. Journal of Econometrics, 68(1), 29-51. https://doi.org/10.1016/0304-4076(94)01642-D
- 12. Athanasoglou, P. P., Brissimis, S. N., & Delis, M. D. (2008). Bank-specific, industry-specific macroeconomic determinants of bank profitability. *Journal of International Financial Markets, Institutions and Money, 18*(2), 121–136. https://doi.org/10.1016/j.intfin.2006.07.001
- 13. Barth, J. R., Caprio, G., & Levine, R. (2005). Rethinking bank regulation: Till angels govern. Cambridge University Press. https://doi.org/10.1017/CBO9780511753817
- Batten, J., & Vo, V. X. (2019). Determinants of bank profitability Evidence from Vietnam. Emerging Markets 14. Finance and Trade, 55(6), 1417-1428. https://doi.org/10.1080/1540496X.2018.1524326
- 15. Berger, A. N., & Bouwman, C. H. (2009). Bank liquidity creation. The Review of Financial Studies, 22(9), 3779-3837. https://doi.org/10.1093/rfs/hhn104
- 16. Berger, A. N., & Bouwman, C. H. S. (2013). How does capital affect bank performance during financial crises? Journal of Financial Economics, 109(1), 146-176. https://doi.org/10.1016/j.jfineco.2013.02.008
- 17. Berlin, M. (2012). Banks and markets: Substitutes, complements, or both? Business Review, Q2, 1-10. https://fraser.stlouisfed.org/title/business-review-federal-reserve-bank-philadelphia-5580/second-quarter-2012-557761/banks-markets-522495
- 18. Blundell, R., & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. Journal of Econometrics, 87(1), 115–143. https://doi.org/10.1016/S0304-4076(98)00009-8
- Boitan, I. A. (2015). Determinants of sustainable banks' profitability. Evidence from EU countries. Financial 19. Studies, 19(1) 21-39. http://fs.icfm.ro/vol19i1p21-39.pdf
- Bolarinwa, S. T., Obembe, O. B., & Olaniyi, C. (2019). Re-examining the determinants of bank profitability in 20. Nigeria. Journal of Economic Studies, 46(3), 633–651. https://doi.org/10.1108/JES-09-2017-0246
- 21. Bolt, W., de Haan, L., Hoeberichts, M., Van Oordt, M. R. C., & Swank, J. (2012). Bank profitability during recessions. *Journal of Banking & Finance, 36*(9), 2552–2564. https://doi.org/10.1016/j.jbankfin.2012.05.011
- 22. Bordeleau, E., & Graham, C. (2010). The impact of liquidity on bank profitability (Working Paper No. 2010-38). Bank of Canada. https://www.bankofcanada.ca/wp-content/uploads/2010/12/wp10-38.pdf
- Buchak, G., Hu, J., & Wei, S.-J. (2021). FinTech as a financial liberator (NBER Working Paper No. 29448). National 23. Bureau of Economic Research. https://doi.org/10.3386/w29448
- Combey, A., & Togbenou, A. (2017). The bank sector performance and macroeconomics environment: Empirical 24. evidence in Togo. International Journal of Economics and Finance, 9(2), 180-188. https://doi.org/10.5539 /ijef.v9n2p180

VIRTUS 321

- 25. Dang, V. D., & Nguyen, H. C. (2022). Bank profitability under uncertainty. *The Quarterly Review of Economics and Finance, 83*, 119–134. https://doi.org/10.1016/j.qref.2021.12.001
- 26. Demirgüç-Kunt, A., & Huizinga, H. (1999). Determinants of commercial bank interest margins and profitability: Some international evidence. *The World Bank Economic Review*, *13*(2), 379–408. https://doi.org/10.1093/wber /13.2.379
- 27. Deng, A. (2016). Determinants and differences of domestic and foreign commercial bank profitability: Bank specific and macroeconomic determinants of large bank profitability in the United States and seven foreign countries (2004-2014) [Honors Theses and Capstones]. University of New Hampshire Scholars' Repository. https://scholars.unh.edu/cgi/viewcontent.cgi?article=1287&context=honors
- 28. DeYoung, R., & Rice, T. (2004). How do banks make money? The fallacies of fee income. *Economic Perspectives*, *28*(4), 34–51. https://www.chicagofed.org/publications/economic-perspectives/2004/4qtr2004-part3-deyoung-rice
- 29. Dietrich, A., & Wanzenried, G. (2011). Determinants of bank profitability before and during the crisis: Evidence from Switzerland. *Journal of International Financial Markets, Institutions and Money, 21*(3), 307-327. https://doi.org/10.1016/j.intfin.2010.11.002
- Farooq, M., Khan, S., Atique Siddiqui, A., Tariq Khan, M., & Kamran Khan, M. (2021). Determinants of profitability: A case of commercial banks in Pakistan. *Humanities and Social Sciences Reviews*, 9(2), 1–13. https://doi.org/10.18510/hssr.2021.921
- 31. Flamini, V., Schumacher, L., & McDonald, C. (2009). *The determinants of commercial bank profitability in Sub-Saharan Africa* (IMF Working Paper No. 09/15). International Monetary Fund (IMF). https://doi.org/10.2139 /ssrn.1356442
- 32. Fungáčová, Z., Klein, P.-O., & Weill, L. (2020). Persistent and transient inefficiency: Explaining the low efficiency of Chinese big banks. *China Economic Review*, *59*, Article 101368. https://doi.org/10.1016 /j.chieco.2019.101368
- 33. García-Herrero, A., Gavilá, A., & Santabárbara, D. (2009). What explains the low profitability of Chinese Banks? *Journal of Banking & Finance, 33*(11), 2080–2092. https://doi.org/10.1016/j.jbankfin.2009.05.005
- 34. Goddard, J., Liu, H., Molyneux, P., & Wilson, J. O. S. (2013). Do bank profits converge? *European Financial Management, 19*(2), 345-365. https://doi.org/10.1111/j.1468-036X.2010.00578.x
- 35. Gugler, K., & Peev, E. (2018). The persistence of profits in banking: An international comparison. *Applied Economics*, *50*(55), 5996–6009. https://doi.org/10.1080/00036846.2018.1489111
- 36. International Monetary Fund (IMF). (2022). *Žimbabwe: 2022 Article IV consultation-press release; staff report; and statement by the executive director for Zimbabwe* (IMF Country Report No. 22/112). https://doi.org/10.5089/9798400209086.002
- 37. Isayas, Y. N. (2022). Determinants of banks' profitability: Empirical evidence from banks in Ethiopia. *Cogent Economics & Finance*, *10*(1), Article 2031433. https://doi.org/10.1080/23322039.2022.2031433
- 38. Islam, M. S., & Nishiyama, S.-I. (2016). The determinants of bank net interest margins: A panel evidence from South Asian countries. *Research in International Business and Finance, 37*, 501–514. https://doi.org/10.1016/j.ribaf.2016.01.024
- Jadah, H. M., Hassan, A. A., Hameed, T. M., & Al-Husainy, N. H. M. (2020). The impact of the capital structure on Iraqi banks' performance. *Investment Management & Financial Innovations*, 17(3), 122–132. https://doi.org/10.21511/imfi.17(3).2020.10
- 40. Jara-Bertin, M., Arias Moya, J., & Rodriguez Perales, A. (2014). Determinants of bank performance: Evidence for Latin America. *Academia Revista Latinoamericana de Administración, 27*(2), 164-182. https://doi.org/10.1108/ARLA-04-2013-0030
- 41. Karakaya, A., & Er, B. (2013). Noninterest (nonprofit) income and financial performance at Turkish commercial and participation banks. *International Business Research*, *6*(1), 106–117. https://doi.org/10.5539/ibr.v6n1p106
- 42. Kararach, G., & Otieno, R. O. (2016). Economic management in a hyperinflationary environment: The political economy of Zimbabwe, 1980–2008. Oxford University Press. https://doi.org/10.1093/acprof:oso /9780198747505.001.0001
- 43. Kohlscheen, E., Murcia, A., & Contreras, J. (2018). *Determinants of bank profitability in emerging markets* (BIS Working Paper No. 686). Bank for International Settlement. https://www.bis.org/publ/work686.pdf
- 44. Komal, R., & Abbas, F. (2015). Linking financial development, economic growth and energy consumption in Pakistan. *Renewable and Sustainable Energy Reviews*, 44, 211–220. https://doi.org/10.1016/j.rser.2014.12.015
- 45. Kosmidou, K. (2008). The determinants of banks' profits in Greece during the period of EU financial integration. *Managerial Finance, 34*(3), 146–159. https://doi.org/10.1108/03074350810848036
- 46. Li, M., & Feng, T. (2016). Relationship between noninterest income and bank valuation: Evidence from the U.S. bank holding companies [Master's thesis, Simon Fraser University]. Simon Fraser University. https://core.ac.uk /download/pdf/80537457.pdf
- Mahonye, N., & Zengeni, T. (2019). Exchange rate impact on output and inflation: A historical perspective from Zimbabwe. *African Journal of Science, Technology, Innovation and Development, 11*(3), 347–358. https://doi.org /10.1080/20421338.2019.1575539
- 48. Mashamba, T. (2018). The effects of Basel III liquidity regulations on banks' profitability. *Journal of Governance* & *Regulation*, 7(2), 34–48. https://doi.org/10.22495/jgr_v7_i2_p4
- 49. Maumbe, B. M., & Chikoko, L. (2022). The rise of supermarkets in Zimbabwe against a tide of macroeconomic uncertainty. *Development Southern Africa*, *39*(3), 289–302. https://doi.org/10.1080/0376835X.2020.1832443
- 50. Maune, A. (2015). Corporate governance in Zimbabwe: An overview of its current state. *Asian Economic and Financial Review*, *5*(1), 167–178. https://doi.org/10.18488/journal.aefr/2015.5.1/102.1.167.178
- 51. Maune, A., Matanda, E., & Mundonde, J. (2020). Is money supply the cause of inflation in Zimbabwe? An empirical examination. *Acta Universitatis Danubius. OEconomica*, *16*(3), 17–37. https://dj.univdanubius.ro/index.php/AUDOE/article/view/372
- 52. Moyo, D., & Tursoy, T. (2020). Impact of inflation and exchange rate on the financial performance of commercial banks in South Africa (MPRA Paper No. 101383). Munich Personal RePEc Archive. https://mpra.ub.uni-muenchen.de/101383/1/MPRA_paper_101383.pdf

VIRTUS

- 53. Muriu, P. (2023). *What explains provisioning behaviour in the banking industry? Evidence from an emerging economy* (Policy Brief No. 78). African Economic Research Consortium. https://publication.aercafricalibrary.org /handle/123456789/3658
- 54. Mutsonziwa, K., & Maposa, O. K. (2016). Mobile money A catalyst for financial inclusion in developing economies: A case study of Zimbabwe using FinScope survey data. *International Journal of Financial Management*, *6*(3), 45–56. http://surl.li/mupps
- 55. Nguyen, D. T., & Le, T. D. (2022). The interrelationships between bank profitability, bank stability and loan growth in Southeast Asia. *Cogent Business & Management, 9*(1), Article 2084977. https://doi.org/10.1080 /23311975.2022.2084977
- 56. Nisar, S., Peng, K., Wang, S., & Ashraf, B. N. (2018). The impact of revenue diversification on bank profitability and stability: Empirical evidence from South Asian countries. *International Journal of Financial Studies, 6*(2), Article 40. https://doi.org/10.3390/ijfs6020040
- 57. Ogunmuyiwa, M. S., & Ekone, A. F. (2010). Money supply-economic growth nexus in Nigeria. *Journal of Social Sciences*, *22*(3), 199–204. https://doi.org/10.1080/09718923.2010.11892802
- 58. Opoku-Mensah, E., Nyakum, J. H., NyankomTakyi, L., & Ampofo, S. A. (2019) Re-examining the impact of credit risk on profitability of banks: Panel evidence from Ghana. *European Journal of Business and Management, 11*(5), 19–24.
- 59. Ozili, P. K. (2015). Determinants of bank profitability and Basel capital regulation: Empirical evidence from Nigeria. *Research Journal of Finance and Accounting*, *6*(2), 124–131. https://doi.org/10.2139/ssrn.2544647
- 60. Ozili, P. K. (2017). Bank profitability and capital regulation: Evidence from listed and non-listed banks in Africa. *Journal of African Business*, *18*(2), 143–168. https://doi.org/10.1080/15228916.2017.1247329
- 61. Philippon, T. (2016). *The fintech opportunity* (NBER Working Paper No. 22476). National Bureau of Economic Research. https://doi.org/10.3386/w22476
- 62. Pracoyo, A., & Imani, A. (2018). Analysis of the effect of capital, credit risk, and liquidity risk on profitability in banks. *Jurnal Ilmu Manajemen & Ekonomika*, *10*(2), 44–50. https://doi.org/10.35384/jime.v10i2.80
- 63. Rao, K., & Lakew, T. B. (2012). Determinants of profitability of commercial banks in a developing country: Evidence from Ethiopia. *International Journal of Accounting and Financial Management Research*, 2(3), 1–20. http://surl.li/mujrb
- 64. Roodman, D. (2006). How to do Xtabond2: An introduction to difference and system GMM in Stata. *The Stata Journal*, *9*(1), 86–136. https://doi.org/10.1177/1536867X0900900106
- 65. Saleh, I., & Abu Afifa, M. (2020). The effect of credit risk, liquidity risk and bank capital on bank profitability: Evidence from an emerging market. *Cogent Economics & Finance, 8*(1), Article 1814509. https://doi.org/10.1080 /23322039.2020.1814509
- 66. Satria, D. M., Harun, C. A., & Taruna, A. A. (2016). The macro-prudential aspects of loan-to-deposit-ratio-linked reserve requirement. *Applied Economics*, *48*(1), 24–34. https://doi.org/10.1080/00036846.2015.1073840
- 67. Scott, A. O., & Ovuefeyen, E. J. (2014). Effects of economic openness and inflation on commercial banks' profitability: Panel data evidence from Nigeria, post-banking sector consolidation (2005–2012). *European Journal of Business and Management, 6*(30), 86–96. https://core.ac.uk/download/pdf/234625841.pdf
- 68. Siddique, A., Khan, M. A., & Khan, Z. (2021). The effect of credit risk management and bank-specific factors on the financial performance of the South Asian commercial banks. *Asian Journal of Accounting Research*, *7*(2), 182–194. https://doi.org/10.1108/AJAR-08-2020-0071
- 69. Sieroń, A. (2019). Endogenous versus exogenous money: Does the debate really matter? *Research in Economics*, *73*(4), 329–338. https://doi.org/10.1016/j.rie.2019.10.003
- 70. Simiyu, C. N., & Ngile, L. (2015). Effect of macroeconomic variables on profitability of commercial banks listed in the Nairobi securities exchange. *International Journal of Economics, Commerce and Management, 3*(4), 1–16. https://ijecm.co.uk/wp-content/uploads/2015/04/3411.pdf
- 71. Singh, R., Malik, G., & Jain, V. (2021). FinTech effect: Measuring impact of FinTech adoption on banks' profitability. *International Journal of Management Practice, 14*(4), 411-427. https://doi.org/10.1504/IJMP.2021.116587
- 72. Subhanij, T. (2010). *Liquidity management and measurement in the SEACEN economies.* The South East Asian Central Banks (SEACEN) Centre. https://core.ac.uk/download/pdf/6281338.pdf
- 73. Sufian, F., & Habibullah, M. S. (2009) Determinants of bank profitability in a developing economy: Empirical evidence from Bangladesh. *Journal of Business Economics and Management*, 10(3), 207–217. https://doi.org/10.3846/1611-1699.2009.10.207-217
- 74. Tan, Y., & Floros, C. (2012). Bank profitability and inflation: The case of China. *Journal of Economic Studies*, 39(6), 675-696. https://doi.org/10.1108/01443581211274610
- 75. Tarus, D. K., Chekol, Y. B., & Mutwol, M. (2012). Determinants of net interest margins of commercial banks in Kenya: A panel study. *Procedia Economics and Finance, 2*, 199–208. https://doi.org/10.1016/S2212-5671(12)00080-9
- 76. Thanh, B. D., Thach, N. N., & Tuan, T. A. (2022). Determinants of bank profitability in Vietnam. In N. Ngoc Thach, V. Kreinovich, D. T. Ha, & N. D. Trung (Eds.), *Financial econometrics: Bayesian analysis, quantum uncertainty, and related topics* (Studies in systems, decision and control: Vol. 427, pp. 517–529). Springer. https://doi.org/10.1007/978-3-030-98689-6_34
- 77. Tran, V. T., Lin, C.-T., & Nguyen, H. (2016). Liquidity creation, regulatory capital, and bank profitability. *International Review of Financial Analysis, 48*, 98–109. https://doi.org/10.1016/j.irfa.2016.09.010
- 78. Yarovaya, L., Mirza, N., Rizvi, S. K. A., & Naqvi, B. (2020). COVID-19 pandemic and stress testing the eurozone credit portfolios. https://doi.org/10.2139/ssrn.3705474
- 79. Yüksel, S., Mukhtarov, S., Mammadov, E, & Özsarı, M. (2018). Determinants of profitability in the banking sector: An analysis of post-Soviet countries. *Economies, 6*(3), Article 41. https://doi.org/10.3390/economies6030041
- 80. Zhang, P., Zhang, M., Zhou, Q., & Zaidi, S. A. H. (2022). The relationship among financial inclusion, nonperforming loans, and economic growth: Insights from OECD Countries. *Frontiers in Psychology*, *13*, Article 939426. https://doi.org/10.3389/fpsyg.2022.939426

VIRTUS